Author's response to reviews

Title: Demographic, socio-economic, and cultural factors affecting fertility differentials in Nepal

Authors:

Ramesh Adhikari (rameshipsr@gmail.com)

Version: 5 Date: 12 April 2010

Author's response to reviews:

Dear Editor,

Please find attached the article entitled "Demographic, socio-economic, and cultural factors affecting fertility differentials in Nepal"

Comments from associate editor has been considered and incorporated in the text accordingly with tracked change. The revised manuscript was edited by a native English speaker (professional English editor). I have corrected the reference. I am confused to make some journal title in abbreviated format so I have written the title of the journal in full.

I trust that the revisions meet the requirements for publication.

Yours sincerely,

Ramesh